Author's response to reviews

Title: Down-regulation of GRP78 is associated with the sensitivity of chemotherapy to VP-16 in small cell lung cancer NCI-H446 cells

Authors:

Yingyan Wang (wyydmu@yahoo.com.cn)
Wei Wang (weiwee1117@yahoo.com.cn)
Siyan Wang (tina2000wan@hotmail.com)
Jiarui Wang (wangjiarui1980@126.com)
ShuJuan Shao (shaoshujuan@sohu.com)
Qi Wang (wqdmu@yahoo.com.cn)

Version: 6 Date: 11 December 2008

Author's response to reviews:

Cover letter

Manuscript ID 3112636332120043

Rebecca Capone
Biotechnology and Bioengineering
Editorial Receiving Office
John Wiley & Sons, Inc.
111 River Street
Hoboken, New Jersey 07030
11, Dec, 2008

Dear Editor:

Thank you very much for the opportunity to revise our manuscript once again “Down-regulation of GRP78 is associated with the sensitivity of chemotherapy to VP-16 in small cell lung cancer NCI-H446 cells (MS: 3112636332120043).

According to your request, we have included the tables in the manuscript and remove the additional files. Please forgive us for misunderstanding the meaning of the format. We upload it now, and if there is still something unsuitable, please do not hesitate to let us know.

Thanks again!

Qi Wang MD PhD
Department of Respiratory Medicine
The Second Hospital
Dalian Medical University
Dalian 116044, China
Tel: 86-0411-86110151
Fax: 86-0411-86110151
Email: wqdmu@yahoo.com.cn